Author's response to reviews

Title: Dissemination of clonal complex 2 Acinetobacter baumannii strains co-producing carbapenemases and 16S rRNA methylase ArmA in Vietnam

Authors:

Tatsuya Tada (ttada@ri.ncgm.go.jp)
Tohru Miyoshi-Akiyama (takiyam@ri.ncgm.go.jp)
Kayo Shimada (kshima@ri.ncgm.go.jp)
Tran T.T Nga (ngatranrch@gmail.com)
Le T.A Thu (letathu@yahoo.com)
Nguyen T Son (truongson_cr@yahoo.com.vn)
Norio Ohmagari (nohmagari@hosp.ncgm.go.jp)
Teruo Kirikae (tkirikae@ri.ncgm.go.jp)

Version: 4
Date: 30 September 2015

Author's response to reviews: see over
Dr. Paul Higgins
Editor for BMC

Teruo Kirikae, M.D., Ph.D.
Director
Department of Infectious Diseases
Research Institute
National Center for Global Health and Medicine
E-mail: tkirikae@ri.ncgm.go.jp

Dear Dr. Paul Higgins:

Thank you for your comments on our manuscript (MS: 1972514606167110). The sections pointed out by the comments are marked in red in the revised manuscript. Our point-by-point responses to the comments are provided in the revised manuscript, and the modified manuscript has been submitted via http://www.biomedcentral.com/manuscript/login/man.asp?txt_nav=man&txt_man_id=1972514606167110.

Yours sincerely,

Teruo Kirikae, M.D., Ph.D.